Math 144 Fall 2006 R. Schultz

UPDATED GENERAL INFORMATION — DECEMBER 4, 2006

Results for the second midterm examination are discussed on the next page.

Further information on the final examination next Tuesday and additional office hours next Monday will be forthcoming. The examination will be about twice as long as the midterm examinations, but there will be three hours in which to complete the examination.

Course evaluations

All course evaluations for both lecture and discussion sections will be done online, and students should receive information on how to complete this via electronic communications. Pleas note that THERE ARE SEPARATE EVALUATIONS FOR THE PRIMARY INSTRUCTOR AND THE DISCUSSION SECTION LEADER, and these are independent of each other. These evaluations are to be completed by the final date of classes (Friday, December 8).

STATISTICS FROM THE SECOND MIDTERM EXAMINATION

The cutoff scores are as follows: A — 90 B — 75 C — 55 D — 35

The median score was 88.

Appeals regarding the grading of this examination must be submitted by the start of the final examination on Tuesday, December 12. Written comments should be placed on the examination indication the problems to be reconsidered. BRIEF and OBJECTIVE statements about specific issues may be included.